Author's response to reviews

Title: Outcome of bone defect reconstruction with clavicle bone cement prosthesis after tumor resection: a case series study

Authors:

- Bin Lin (hy1232@126.com)
- Yong He (hy1232@126.com)
- Yang Xu (hy1232@126.com)
- Mo Sha (hy1232@hotmail.com)

Version: 9 Date: 17 April 2014

Author's response to reviews: see over
Author's response to reviews

Title: Outcome of bone defect reconstruction with clavicle bone cement prosthesis after tumor resection: a case series study

Authors:
BinLin (hy1232@126.com)
YongHe (hy1232@126.com)
YangXu (hy1232@126.com)
MoSha (hy1232@hotmail.com)

Version: 5
Date: 20 December 2013

Author's response to reviews: see over

The Biomed Central Editorial Team

Object: MS: 1304435312112158 - Outcome of bone defect reconstruction with clavicle bone cement prosthesis after tumor resection: a case series study. Dr BinLin et al.

Thank you for consideration of our manuscript for publication in your journal. We have reviewed the above manuscript according to your comments.

Please ensure that the full name of the institutional review board that approved the study is indicated in the manuscript.
We also need clarification whether written informed patient consent was obtained.
The table contained in the additional files section contains sensitive patient information and there are also patient images in the manuscript. Please confirm whether written informed consent was obtained from the patients for publication of their images and data. Please ensure that this is made clear in the manuscript.
Please also include legends for all figures, table and additional file.

It has been changed as the reviewer indicates. (page4, line15-18)